Author’s response to reviews

Title: No changes in levels of bone formation and resorption markers following a broad-spectrum antibiotic course

Authors:
Kristian Mikkelsen (kristian.hallundbaek.mikkelsen@regionh.dk)
Tina Vilsbøll (t.vilsboll@dadlnet.dk)
Jens Holst (jjholst@sund.ku.dk)
Bolette Hartmann Hartmann (bhartmann@sund.ku.dk)
Filip Knop (filipknop@dadlnet.dk)
Morten Frost (frostnielsen@yahoo.com)

Version: 2 Date: 03 Aug 2018

Author’s response to reviews:

Dear Editor,

As suggested, we have revised the manuscript, which includes a list of abbreviations, details on funding, and a combination of figures 1 and 2 (presented as figure 1). A clean version of the manuscript has been uploaded as requested. We do hope that these minor changes are acceptable.

Please note that some of the co-authors has additional affiliations, which could not be included due to some technical issues. The affiliations in the online system, i.e. the "Add/Edit/remove Authors" section, are correct, and all the affiliations are available in the manuscript.

Yours sincerely,
Morten Frost